Author's response to reviews

Title: Temporal and Geographic variation in the validity and internal consistency of the Nursing Home Resident Assessment Minimum Data Set 2.0

Authors:

Vincent Mor (Vincent_Mor@brown.edu)
Orna Intrator (Orna_Intrator@brown.edu)
Mark A Unruh (Mark_Unruh@brown.edu)
Shubing Cai (Shubing_Cai@brown.edu)

Version: 5 Date: 30 March 2011

Author's response to reviews:

I have edited the contributions section to include only initials, added the affiliation separately for each author on the title page and divided the additional files into two different files consistent with the text of the document.